



# Article LungNet22: A Fine-Tuned Model for Multiclass Classification and Prediction of Lung Disease Using X-ray Images

F. M. Javed Mehedi Shamrat <sup>1</sup>, Sami Azam <sup>2</sup>,\*<sup>1</sup>, Asif Karim <sup>2</sup>, Rakibul Islam <sup>3</sup>, Zarrin Tasnim <sup>1</sup>, Pronab Ghosh <sup>4</sup> and Friso De Boer <sup>2</sup>

- <sup>1</sup> Department of Software Engineering, Daffodil International University, Dhaka 1207, Bangladesh; javedmehedicom@gmail.com (F.M.J.M.S.); zarrint25@gmail.com (Z.T.)
- <sup>2</sup> College of Engineering, IT and Environment, Charles Darwin University, Casuarina, NT 0909, Australia; asif.karim@cdu.edu.au (A.K.); friso.deboer@cdu.edu.au (F.D.B.)
- <sup>3</sup> Department of Computer Science and Engineering, Daffodil International University, Dhaka 1207, Bangladesh; rakibulislam9200.cse@gmail.com
- <sup>4</sup> Department of Computer Science (CS), Lakehead University, 955 Oliver Rd, Thunder Bay, ON P7B 5E1, Canada; pghosh1@lakeheadu.ca
- \* Correspondence: sami.azam@cdu.edu.au

Abstract: In recent years, lung disease has increased manyfold, causing millions of casualties annually. To combat the crisis, an efficient, reliable, and affordable lung disease diagnosis technique has become indispensable. In this study, a multiclass classification of lung disease from frontal chest X-ray imaging using a fine-tuned CNN model is proposed. The classification is conducted on 10 disease classes of the lungs, namely COVID-19, Effusion, Tuberculosis, Pneumonia, Lung Opacity, Mass, Nodule, Pneumothorax, and Pulmonary Fibrosis, along with the Normal class. The dataset is a collective dataset gathered from multiple sources. After pre-processing and balancing the dataset with eight augmentation techniques, a total of 80,000 X-ray images were fed to the model for classification purposes. Initially, eight pre-trained CNN models, AlexNet, GoogLeNet, InceptionV3, MobileNetV2, VGG16, ResNet 50, DenseNet121, and EfficientNetB7, were employed on the dataset. Among these, the VGG16 achieved the highest accuracy at 92.95%. To further improve the classification accuracy, LungNet22 was constructed upon the primary structure of the VGG16 model. An ablation study was used in the work to determine the different hyper-parameters. Using the Adam Optimizer, the proposed model achieved a commendable accuracy of 98.89%. To verify the performance of the model, several performance matrices, including the ROC curve and the AUC values, were computed as well.

Keywords: LungNet22; X-ray image; VGG16; optimizer; CNN; lung disease; deep learning

# 1. Introduction

Lung disease is categorized as the third leading cause of mortality globally [1], with an estimated total of five million deaths per year [2]. As a vital respiratory organ, the lung is not only prone to infection from components but externally as well. Lung disease manifests due to various forms of air pollution, microbiological attack, chemical intake, or physical sickness. One of the most alarming lung diseases in recent times is COVID-19. COVID-19 is detected by identifying upper respiratory tract and lung infection accompanied by pneumonia and a cold-like syndrome. The disease was first explored in Wuhan, China [3]. COVID-19 causes the air sacs in the lungs to fill with fluids leaking from the blood vessels, resulting in difficulty in breathing. Among some very common lung diseases are pneumonia and pneumothorax. Pneumothorax occurs when air escapes between the lung and the chest wall of the patient, causing the lung to completely collapse. On the other hand, pneumonia is an acute respiratory infection that causes the alveoli in the lung to fill with pus or fluid [4]. Abnormal growth in the lung indicates a lung tumor. If the growth is



Citation: Shamrat, F.M.J.M.; Azam, S.; Karim, A.; Islam, R.; Tasnim, Z.; Ghosh, P.; De Boer, F. LungNet22: A Fine-Tuned Model for Multiclass Classification and Prediction of Lung Disease Using X-ray Images. *J. Pers. Med.* 2022, 12, 680. https://doi.org/10.3390/ jpm12050680

Academic Editors: Chin-Sheng Lin, Chin Lin and Hung-Yu Wei

Received: 10 March 2022 Accepted: 20 April 2022 Published: 24 April 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). larger than 3 cm, it is called a mass. However, if the growth is less than 3 cm in diameter, it is known as a nodule. A benign nodule is not cancerous; however, a malignant nodule not only spreads to other parts of the body but also causes lung cancer. Effusion refers to a build-up of fluid. An excess amount of fluid builds up outside the lung and within the pleural layers [5]. Pulmonary fibrosis occurs due to damage and scars in the lung; thick and stiff tissues grow, making it laborious to breathe. Tuberculosis damages the lung by creating a large cavity in the lung and enlarging bronchiectasis. Lung opacity is identified as an area in the lung damaged by intraparenchymal hemorrhage.

Medical X-ray imaging is a universal diagnosis technology for disease diagnosis. Using X-ray, the internal physical structure of organs and bones can be detected [6]. X-rays are valuable diagnosis techniques that have been utilized for years by experts to diagnose fractures, some cancers, pneumonia, dental issues, and other conditions. In severe circumstances, Computed Tomography (CT) may provide a succession of body scans that are then combined to create a three-dimensional X-ray picture analyzed by a computer. However, a normal X-ray is quicker, simpler, less expensive, and less invasive than a CT scan [7]. In the majority of instances, X-rays are insufficient to provide a diagnosis; CT scans are often required to confirm the diagnosis [8]. Multiple CT scans may be required for diseases that develop rapidly, but this is exceedingly costly, time-demanding, and may potentially be harmful to the patient as both of them use radiation [9].

As a result, there is a critical demand for an artificial intelligence system capable of effectively detecting chest-related illnesses [10,11]. The research includes applying deep learning to create predictions about medical pictures by extracting information such as spatial rotation and shape from the photos. CNNs have been important in the extraction of features and in the learning of patterns that allowed prediction [12]. AI has matured to the point that it can be incorporated with cutting-edge machine learning and deep learning algorithms in a variety of disciplines, including health [7], biometrics [13], agriculture [14], cloud computing [15], and renewable energy [16]. Numerous advantages are associated with AI technology in medicine. For example, rural locations and third-world countries often lack specialized physicians capable of providing the necessary treatment [17]; artificial intelligence technologies support them in obtaining the necessary medical treatment.

Additionally, this kind of research will benefit the healthcare industry significantly as professionals will be able to utilize it to support their diagnoses; additionally, it may be operated by those with no medical training. Even patients might benefit from this technology [18,19]. All of this is done to mitigate the specialist's load, given that there are only a few of them, by focusing them only on the X-rays marked as problematic by the prediction tools. Moreover, they significantly greatly reduce a doctor's subjective decisions, accelerate diagnosis, and reliably identify details that the human vision may miss [20].

In our study, X-ray images of the chest were collected from multiple sources. The datasets were collected from databases such as GitHub, Kaggle, and NIH Clinical Center [21–36]. The collection is composed of multiple lung-disease X-ray images that include COVID-19, Effusion, Lung Opacity, Mass, Nodule, Pulmonary Fibrosis, Pneumonia, Pneumothorax, and Tuberculosis, in addition to the Controlled class. The disease's features may be easily and quickly detected using deep learning models, allowing for a more rapid and accurate diagnosis.

The contributions of the paper are as follows:

- 1. The raw chest X-ray data for the study is collected from multiple data sources, merged into a collective dataset. To ensure all the data are in the same scale, preprocessing is performed.
- 2. Autoencoder is employed to denoise the noisy images. For removing annotation in the data, the EnsNet method is used.
- CLAHE enhancement methods are used to increase the quality of the X-ray images, and the Green Fire Blue filtering technique is applied to improve the image properties and make them more identical to the model.

- 4. Augmentation is performed to balance each class of the dataset by upsampling the classes with fewer data.
- 5. Eight pre-trained models, namely AlexNet, GoogLeNet, InceptionV3, MobileNetV2, VGG16, ResNet 50, DenseNet121, and EfficientNetB7 are evaluated to determine the highest-performing model.
- 6. A fine-tuned LungNet22 is proposed based on VGG16 architecture.
- 7. To justify the model's accuracy, the precision, recall, specificity, and f1-score are calculated from the confusion matrix. The loss function is also used to evaluate the performance of the models.
- 8. For the visual description of the model's outcome, Grad-CAM is produced.
- 9. The proposed model is evaluated by computing the ROC curve and the AUC value. The operating process of the proposed study is presented in Figure 1.



Figure 1. Operation of the proposed LungNet22 model for multiclass classification of lung disease.

The rest of the paper is organized in the following manner. Section 2 summarizes the prior research on lung disease diagnosis in multiple datasets, using deep learning with the models' accuracy scores. Section 3 discusses the collection, pre-processing, and preparation of the datasets prior to feeding the model. Section 4 covers the transfer learning models that were chosen, and their parameters, and the proposed model (LungNet22) with the ablation study by applying different parameters. The result is discussed in Section 5. Section 6 summarizes the study's findings and compares those to other related research. Section 7 highlights the research findings and makes recommendations for further research.

#### 4 of 29

#### 2. Literature Review

The capacity of machine learning algorithms, particularly that of deep learning, to identify anomalies in X-ray images has gained popularity in recent years. Artificial intelligence is being used in medical research to facilitate diagnosis, and some of the studies have shown positive and accurate findings. The strategies employed by past researchers to handle lung-related disorders by utilizing artificial intelligence and deep neural networks are explored here.

A multiclass classification of the deep learning model is suggested in [37] for identifying COVID-19, pneumonia, and lung cancer using chest X-ray and CT images. The efficiency of four architectures, which are a combination of VGG19, ResNet152V2 with CNN, GRU, and Bi-GRU, is examined the accuracy of the VGG19 + CNN model was 98.05%. The authors in [38] present a novel, hybrid deep learning framework called VGG Data STN, with CNN, VDSNet that combines with VGG, data augmentation, and a spatial transformer network (STN) with CNN. The other models implemented in the study are Vanilla Gray, Vanilla RGB, Hybrid CNN + VGG, and a modified Capsule Network. The proposed VDSNet model has a validation accuracy of 73%. This study [39] provided an accurate technique of classifying COVID-19, pneumonia, and the normal class, utilizing the CNN classification method with the histogram-oriented gradient (HOG) feature-extraction methods. A public X-ray dataset has been used to train and test the proposed CNN model. The model is validated by the use of a 10-fold cross-validation and metric confusion.

To classify and forecast chest X-rays for various lung diseases, the authors in [40] recommended utilizing the proposed model UCMobN. The model is constructed by modifying MobileNetV2 architecture. Atelectasis, Consolidation, Edema, Effusion, Emphysema, Fibrosis, Infiltration, and Mass are among the 10 lung disorders included in the dataset. In the study [41], the authors provide a CNN-based model for automatically detecting pleural effusion in chest X-rays. The research relies on X-ray scans of patients with pleural effusion and those in a healthy state. An accuracy of 95% in the classifying of the data was found using the CNN model in this examination. The authors in [42] verify a deep CNN technique termed Decompose, Transfer, and Compose (DeTraC), which is capable of handling any anomalies in the image dataset. The approach employs a variety of CNN models, including AlexNet, VGG19, ResNet, GoogleNet, and SqueezeNet to classify data with an accuracy of 97.53%.

Another article [43] proposes a method for augmenting synthetic data in three deep CNN models in order to diagnose 14 chest-related disorders. DenseNet121, Inception-ResNetV2, and ResNet152V2 are the models used. To identify abnormalities in X-ray scans, the suggested models were trained for multiclass classification. The authors in [44] employed the Mask R-CNN approach on the X-ray dataset in order to categorize patients with and without the COVID-19 infection. Five-fold cross-validation was used to train the Mask R-CNN for 100 epochs. The authors suggest an architecture for the COVID-aid model based on the DarkCovidNet architecture for COVID-19 and pneumonia detection in [45]. This model has 19 convolutional layers and six max-pooling layers. The authors present a CNN-based classifier for pneumonia diagnosis and validate the numbers to choose the optimum model for the job depending on the specific characteristics in [46]. To X-ray images, the research applies the VGG16, VGG19, NasNetMobile, ResNet152V2, and InceptionResNetV2 models.

Another paper [47] makes use of many state-of-the-art CNN models, DenseNet201, ResNet50V2, and InceptionV3 for detecting COVID-19 patients using chest X-ray scans. Initially, the models are separately taught to produce self-contained forecasts and then integrated using weighted average ensembling to predict a class value. In [48], using chest X-ray images as inputs, an automated CNN model is constructed and suggested to classify COVID-19 into four severity classes: COVID-19 (Mild), COVID-19 (Moderate), COVID-19 (Severe), and COVID-19 (Critical). The model produced an average accuracy of 95.52%. Almost all of the hyper-parameters in the CNN are tuned automatically by the grid search optimizer.

The authors in [49] proposed two algorithms for COVID-19 detection, including DNN techniques based on the fractal characteristic of data and CNN methods based on lung images. The classification results indicate that the provided CNN architecture is more accurate (93.2%) than the DNN technique (83.4%). Deep learning techniques, the fine-tuning of pre-trained CNNs, and the end-to-end training of a built CNN model were utilized in this research [50] to detect COVID-19 and normal chest X-rays. Pre-trained deep CNN models were employed for deep feature extraction. An SVM classifier was utilized in conjunction with several kernel functions to classify the deep features. Additionally, the fine-tuning technique employed the previously pre-trained deep CNN models.

The authors of this article [51] develop a Multi-scale Adaptive Residual Neural Network (MARnet) for the purpose of classifying chest X-ray scans of lung illnesses. To improve the model, the authors extract image characteristics and cross-transfer the information retrieved from the residual block and the information extracted from the adaptive structure to a separate layer, avoiding the residual structure's reducing influence on the adaptive function. The accuracy gained by the literature investigations is summarized in Table 1.

Reference	Year	Classes	Disease	Data Type	Model	Performance Accuracy
[37]	2021	4	Normal, COVID-19, Pneumonia, Lung cancer	X-ray, CT	VGG19-CNN ResNet152V2 ResNet152V2 + GRU ResNet152V2 + Bi-GRU	98.05% 95.31% 96.09% 93.36%
[38]	2020	15	No finding, Infiltration, Mass, Effusion, Atelectasis, Nodule, Pneumothorax, Consolidation, Plural thickening, Hernia, Cardiomegaly, Emphysema, Edema, Fibrosis, Pneumonia	X-ray	Vanilla Gray Vanilla RGB Hybrid CNN VGG VDSNet Modified CapsNet Basic CapsNet	67.8% 69% 69.5% 73% 63.8% 60.5%
[39]	2021	3	Normal, Pneumonia, COVID-19	X-ray	CNN + HOG	92.95%
[40]	2020	10	Infiltration, Mass, Effusion, Atelectasis, Nodule, Pneumothorax, Consolidation, Emphysema, Edema, Fibrosis	X-ray	UCMobN	85.62%
[41]	2021	2	Pleural effusion, Normal	X-ray	CNN	95%
[42]	2021	3	Normal, COVID-19, SARS	X-ray	DeTrac + AlexNet DeTrac + VGG19 DeTrac + ResNet DeTrac + GoogleNet DeTrac + SqueezNet	95.66% 97.53% 95.66% 94.71% 94.90%
[43]	2021	14	Atelectasis, Cardiomegaly, Consolidation, Edema, Mass, Effusion, Emphysema, Nodule, Pneumothorax, Fibrosis, Infiltration, Plural thickening, Pneumonia, Normal	X-ray	DenseNet121 InceptionResNetV2 ResNet152V2	40.43% 41.02% 37.60%
[44]	2021	2	COVID-19, Not-COVID-19	X-ray	Mask R-CNN	96.98%
[45]	2021	3	No finding, COVID-19, Pneumonia	X-ray	CNN	87%
[46]	2020	2	Normal, Pneumonia	X-ray	VGG16 VGG19 NasNetMobile ResNet152V2 InceptionResNetV2	96.81% 96.58% 83.37% 96.35% 94.87%
[47]	2021	2	Normal, COVID-19	X-ray	INceptionV3 ResNet50V2 DenseNet201 Combined Proposed Model	90.43% 91.11% 91.11% 91.62%
[48]	2021	4	COVID-19 (Mild), COVID-19 (Moderate), COVID-19 (Severe), COVID-19 (Critical)	X-ray	Novel CNN	95.52%
[49]	2020	2	Normal, COVID-19	X-ray	DNN CNN	83.4% 93.2%
[50]	2021	2	Normal, COVID-19	X-ray	ResNet50 + SVM Fine-Tune ResNet50 End-to-End training of CNN	94.7% 92.6% 91.6%
[51]	2022	4	Nodules, Atelectasis, Infection, Normal	X-ray	MARnet	83.3%

Table 1. A summary of related studies on lung disease classification.

# 3. Datasets

In the study, X-ray images of chests were collected from multiple sources. The datasets were collected from databases such as GitHub, Kaggle, and NIH Clinical Center [21–36]. The collective dataset is a combination of 16 datasets and contains 85,105 frontal chest X-ray images. The collection consists of 10 classes of X-ray images that include COVID-19 (15,660 images), Effusion (13,501 images), Lung Opacity (7179 images), Mass (5603 images), Nodule (6201 images), Pulmonary Fibrosis (3357 images), Pneumonia (9878 images), Pneumothorax (6870 images), and Tuberculosis (3184 images), in addition to the Controlled (13,672 images) class. Table 2 contains the information of the sources of data collected for each class. The age frequency of the dataset is between 1 and 89 years. The data of both male and female patients are included in the dataset. The study is conducted to identify lung disease irrespective of age and gender. A sample of the raw dataset is presented in Figure 2.

Table 2. The raw dataset's description.

No.	Disease	Data	Reference
1.	Control	13,672	[21-23,26-28,31]
2.	COVID-19	15,660	[21,22,24,25,28,31]
3.	Effusion	13,501	[30,32–34]
4.	Lung Opacity	7179	[31,35]
5.	Mass	5603	[30,33]
6.	Nodule	6201	[30,33]
7.	Pulmonary Fibrosis	3357	[28,30,33]
8.	Pneumonia	9878	[21-23,28,31]
9.	Pneumothorax	6870	[29,30,33]
10.	Tuberculosis	3184	[24,26–28,36]



Nodule

Pulmonary Fibrosis

Pneumonia

Tuberculosis

**Figure 2.** Sample dataset of the ten classes, including Control, COVID-19, Effusion, Lung Opacity, Mass, Nodule, Pulmonary Fibrosis, Pneumonia, Pneumothorax, and Tuberculosis.

Pneumothorax

# 3.1. Dataset Preprocessing

The X-ray data are gathered from a variety of sources, due to the fact that the CNN model used for classification needs clean, enhanced, and balanced picture data. To feed the model with a high-quality picture, image-preparation and image-balancing techniques are utilized. This section discusses in depth the image-processing methods.

#### 3.1.1. Image Denoising

Autoencoder is a reasonable solution because of its implementation in denoising, which has a lot of potential for the extraction of features and data-component identification, which are the early stages before entering further into imaging analysis and processing. The Denoising Autoencoder (DAE) acquires the input parameters during picture reconstruction, leading to an overall improvement in latent representational extraction [52].

As part of the training of the model, denoising is a smart option as it preserves input info (input encode) and tries to eliminate noise from the auto-encoder input. The DAEs have been demonstrated to be edge detectors for natural picture patches and bigger stroke detectors for digit images, respectively [53]. Finally, DAEs outperform standard denoising filters because they can be adjusted to the input, while traditional filters are not data-specific. The autoencoder consists of an encoder and a decoder at a high level. "Autoencoder" is a term that refers to the automated functions of these two sections. The encoder reduces the dimensionality of the input signal to a more compressed latent state, while the decoder reverses the encoder's work on the decoded output to recreate the original picture. The latent state of an image cannot be reconstructed by traditional autoencoders. Another variant of this is to delete portions of the input rather than adding noise to the input, in order for the model to learn to forecast the original picture.

The dataset (image size:  $224 \times 224$  pixel) has been used to implement the DAE. The CNN is used in this approach. Due to its efficacy in capturing spatial details, the CNN is the chosen neural network for image dataset analysis. The DAEs procedure begins with the dataset being loaded and the pixel values being normalized. The random noise is then applied to the train and test sets through the function "np.random.noise". Thus, the noisy picture will be utilized as the feed for the encoder and the principal images as the output for training/testing the model. The formula "F(X) = Y" is essentially used, where X is the source noise-free picture, and Y is indeed the noisy image.

During training, a loss function similar to the root mean squared error (RMSE) is set, and the network estimates and tries to decrease the losses among the denoised images (reconstructed pictures) from the decoder and the raw picture in each iteration. From Equation (1), the validation loss is measured by considering the network output ( $y^i$ ) to the actual output ( $y_i$ ), and as the network advances on inputs, this loss may be predicated on a loss function, where  $\mathcal{L}$  is the individual loss calculated by taking into account the difference between the predicted and the actual output.

$$J = 1/N \sum N_i = 1\mathcal{L}\left(y^i, y_i\right) \tag{1}$$

The architecture of a convolutional neural network is characterized as having two key components: an encoder that performs feature extraction and a decoder. Essentially, the encoder scans the picture using filters and increases the depth of the image to allow for greater feature extraction, while the decoder reconstructs the same image. The results after denoising images are exhibited in Figure 3.

#### 3.1.2. Image Annotation Remove

The objective of image text erasing is to enshrine the text and replace it with a visually convincing backdrop while leaving the non-text sections untouched. The following are the difficulties associated with deleting the scene text: (1) image text removing methods should be capable of anticipating the text's stroke-level position, which is more complicated than the bounding-box-level scene text recognition methods that have been extensively researched [54]; (2) after trying to remove words, the source text region must be loaded with a fresh visually plausible background; and (3) non-text regions should retain their original appearance. The EnsNet [55] network is used to remove text from images. It is made of three seamlessly linked modules: (1) the refine generating core; (2) the finely tuned module; and (3) the EnsNet discriminator. The network is end-to-end trainable with an extraordinarily high efficiency.



Figure 3. The output and architecture of the image denoising from raw noisy images.

EnsNet is made up of two fundamental components: a generator G and a discriminator D. EnsNet aims to construct a non-text picture y that is as real as appropriate, given a scene text image x and a ground truth z, by solving the optimization problem [56], which may be described using Equation (2).

$$min_G max_D E_{x \sim p_{data(x)}}, \ z[\log(1 - D(x, G(x, z)))] + E_{x \sim p_{data(x)}}, \ z[\log D(x, y)]$$
 (2)

Here, *G* is composed of three modules that are mutually promoted: (1) a lightweight Fully-Convolutional Networks (FCN) core, (2) lateral connections, and (3) reduced loss.

(1) The FCN is composed of a convolutional and a deconvolutional route. The former makes use of a low-power Resnet18 network. To forecast the text/non-text score map, the  $1 \times 1$  convolutional layer transformed with the last layer is put on top of the Resnet18's final convolutional layer. The deconvolution route is composed of 5 deconvolutional layers, each having a kernel size of 4, a stride step size of 2, and a padding size of 1.

(2) The lateral connections often are thought to have better semantics, while the higher-level features have weaker semantics but include more specific information, such as pixel-level color, texture, and object location information [57]. As a result, they created lateral links to relate higher-level semantics to lower-level features. A transforming block and an upsampling block are included in the proposed lateral connection. The transforming block begins with a shrinking layer that uses a  $1 \times 1$  convolution to minimize the feature dimensions. Following that, two convolutional layers of the same size  $(3 \times 3)$  are stacked to provide a nonlinear transformation that not only replaces the sizable convolution layer [58] but also improves the processing efficiency. Finally, an expanding layer is utilized to reverse the shrinking step by enlarging the convolutional feature channels using a  $1 \times 1$ convolution. The transformation block accepts as input the Residual2b to Residual5b of the Resnet18. In addition, a deconvolutional layer is employed to increase the feature map in the upsampling block [59]. The upsampled features maps are then added elementby-element to the converting block's corresponding ones. Additionally, unlike earlier CNN-based approaches, all ReLU/LeakyReLU levels, except those in the convolution route, are replaced with ELU layers after each layer.

(3) Refined loss algorithms take into account both per-pixel reconstruction accuracy and composition, such as how well text sections blend into their surrounding context. This generative network is equipped with four functions: (i) loss of multiscale regression, (ii) loss of content, (iii) loss of texture, and (iv) loss of total variation. In the multiscale regression loss, an input picture containing text  $I_{in}$ , an initial binary mask M, a generator output  $I_{out}$ , a ground truth image  $I_{gt}$ , and extracted features from several deconvolutional layers are provided to generate outputs of varying sizes. This allows for the gathering of additional contextual information at various sizes, where  $I_{out(i)}$  denotes the *ith* deconvolutional output,  $M_i$  and  $I_{gt(i)}$  denote the segmentation masks and ground truth with the same scale as  $I_{out(i)}$ , and weights are given of the relevance of the text and non-text areas. The weight of the *ith* scale is denoted by  $\lambda_i$ . The loss function for the multiscale regression is defined in Equation (3).

$$\mathcal{L}_m(M, I_{out}, I_{gt}) = \sum_{i=1}^n \lambda_i \Big( \parallel M_i \odot \Big( I_{out(i)} - I_{gt(i)} \Big) \parallel 1 + \alpha \parallel (1 - M_i) \odot \Big( I_{out(i)} - I_{gt(i)} \Big) \parallel 1 \Big)$$
(3)

The content loss function has been shown to be effective for recreating high-level features [60]. To further improve the speed of text erasing, also included are the content limits on the high-level characteristics, as described in Equation (4); it refers to them as content loss, where  $I_{comp}$  is the output image  $I_{out}$ , with  $I_{out}$ 's non-text areas set towards the ground truth. The activation map for the *nth* chosen layer is denoted by  $A_n$ .

$$\mathcal{L}_{C} = \sum_{n=1}^{N-1} ||(A_{n}(I_{out}) - A_{n}(I_{gt}))||1) + \sum_{n=1}^{N-1} ||(A_{n}(I_{comp}) - A_{n}(I_{gt}))||1)$$
(4)

The texture loss should be included in the optimization process as well. As a result, texture degradation is introduced, ensuring that the recovered text portions match the non-text sections. The current success of neuronal style transfers [61] has prompted the loss. Before performing the *L*1 loss, which may be described by Equations (5) and (6), texture loss conducts the autocorrelation [62] on every high-level feature map. When  $(H_n W_n) \times C_n$  denotes the form of the high-level activation map  $(A_n)$ , the discrepancy between the texture appearances of the text and non-text areas is penalized so that the network may collect the global style characteristics for more fairly shifting the textual areas. Again, loss terms are included for both the unprocessed result *I*<sub>out</sub> and the text-erased areas' output *I*<sub>comp</sub>.

$$\mathcal{L}_{T_{out}} = \sum_{n=1}^{N-1} || \frac{1}{C_n H_n W_n} ((A_n(I_{out}))^T (A_n(I_{out})) - ((A_n(I_{gt}))^T (A_n(I_{gt}))) || 1$$
(5)

$$\mathcal{L}_{T_{comp}} = \sum_{n=1}^{N-1} || \frac{1}{C_n H_n W_n} ((A_n (I_{comp}))^T (A_n (I_{comp})) - ((A_n (I_{gt}))^T (A_n (I_{gt})) || 1$$
(6)

The loss of total variation  $L_{tv}$  is intended to achieve global denoising, as specified in Equations (7) and (8), where *i* and *j* denote the pixel dimensions. In Equation (8), the hyper-parameters,  $\lambda_e$ ,  $\lambda_i$ , and  $\lambda_t$ , determine the balance of the four losses.

$$\mathcal{L}_{tv} = \sum_{i,j} ||I_{out}^{i,j} - I_{out}^{i+1,j}||1 + ||I_{out}^{i,j} - I_{out}^{i,j+1}||1$$
(7)

$$\mathcal{L}_{refined} = \mathcal{L}_M + \lambda_e \mathcal{L}_c + \lambda_i \mathcal{L}_T + \lambda_t \mathcal{L}_{tv} \tag{8}$$

To distinguish between false and genuine photos, the initial GANs [63] differentiate the results at the entire image level. However, as the non-text portions comprise a large amount of the picture and are often genuine, it is difficult for the discriminator to concentrate on the text sections. As a result, it is natural to build a local-aware discriminator that is conscientious about the consistency between the text-erased patches and their underlying texture. The output after applying the image annotation method is (EnsNet) illustrated in Figure 4. In this figure, the annotations marked in red circles.



Figure 4. The output after deploying the image annotation removal method.

# 3.1.3. Image Enhancement (CLAHE)

To enhance the contrast of an image, Contrast Limited Adaptive Histogram Equalization (CLAHE) is utilized. CLAHE is a more advanced version of Adaptive Histogram Equalization (AHE). The research in [64] examined the two approaches and stated why CLAHE is preferable to AHE. CLAHE was developed to improve the quality of the imaging of complex structures in medicine [65]. CLAHE improves the local contrast of medical imaging and its usability [66].

CLAHE has been shown to be effective at enhancing low-contrast pictures [67]. The CLAHE method divides the pictures into contextual sections known as tiles and generates a histogram of each one and approximates the result to a given histogram distribution parameter. The study in [68] provided a concise description of the mathematical formulas regulating CLAHE. Imagine that a picture is  $N \times N$  pixels in size and each tile in the image is  $n \times n$  pixels in size. The overall quantity of tiles is then calculated using Equation (9),

$$T = \frac{N \times N}{n \times n} \tag{9}$$

The necessary histograms of the above tiles are constructed by exploiting the clip limit,  $C_L$  as specified in Equation (10).

$$C_L = N_{CL} \times N_{AVG} \tag{10}$$

Here,  $N_{CL}$  stands for Normalized Contrast Limit.  $N_{AVG}$  is the average count of pixels in the picture. The ratio of  $N_{AVG}$  is determined using Equation (11).

$$N_{AVG} = \frac{N_x \times N_y}{N_g} \tag{11}$$

Here,  $N_g$ ,  $N_x$ , and  $N_y$  in the equation represent the number of grayscales, the pixel size in the x dimension, and the pixel size in the y dimension, respectively, in the tile. The number of mean values of clip pixels is calculated using Equation (12).

$$N_{cp} = \frac{N \sum C_L}{N_g} \tag{12}$$

Here,  $C_L$  stands for clipped pixels, while  $N \sum C_L$  stands for the overall number of  $C_L$ . Some pixels remain after the  $N_{cp}$  is spread across the various gray levels. It is used to redistribute the leftover pixels, using Equation (13).

$$R = \frac{N_g}{N_r} \tag{13}$$

where *R* is calculated by dividing  $N_g$  by the image pixels to be reallocated,  $N_r$ . To reduce false borders in the image data, bi-linear interpolation is used to combine neighboring tiles.

In order to overcome the limitations of global approaches, the CLAHE methodology focuses on increasing local contrast. The size of the tiles and the clip limit are key hyperparameters for this approach. An erroneous selection of hyper-parameters may have a significant impact on the quality of the picture. Numerous values for these parameters are evaluated, and the ones that provide the best results are picked for tileGridSize (10, 10), the clip limit (3). An image histogram of the CLEHE and the raw pictures is created to indicate the intensity levels of the image pixels. A statistical representation of the pictures is used in the study, with an intensity range of pixel 0–255. As seen in the histogram, the contrast of a CLAHE picture is significantly superior to that of the original raw image. The output results after applying CLEHE are shown in Figure 5.



**Figure 5.** The result of CLAHE enhancement method implemented on ten classes: Control, COVID-19, Effusion, Lung Opacity, Mass, Nodule, Pulmonary Fibrosis, Pneumonia, Pneumothorax, and Tuber-culosis, applying several tileGridSize (10, 10) and clip limit (3) permutations.

#### 3.1.4. Filter Apply

The last enhancing technique in this research is to apply a filter. The fundamental concept of filtering is that single pixels in a picture are given a new value based on the values of the adjacent pixels within a specific zone. Different filters generate their output by doing various computations on the neighborhood. This filtering technique is used to enhance the image's properties and render it more identical to the model for training, validation and testing. As shown in Figure 6, multiple filters (5 Ramps, 3-3-2 RGB, 6 shades, 16 colors, Blue Orange icb, Spectrum, Fire, Sepia, Cyan Hot, 16 colors, Edges, Phase, Jet, Green Fire Blue, Thallium, and Viridis) are applied to the X-ray pictures, and the optimum



filter with the excellent accuracy is chosen. The improved picture is given as output in Figure 7, following the application of the CLAHE procedures.

**Figure 6.** The result of different filtering algorithms includes the following: 5 Ramps, 6 Shades, 3-3-2 RGB, Blue Orange icb, 16 colors, Spectrum, Sepia, Cyan Hot, Edges, Fire, Green Fire Blue, Phase, Jet, Thallium, and Viridis were applied to the X-ray image dataset.

As seen in Figure 6, the lung damage is more easily recognized with Green Fire Blue (GFB) than with the other filters. However, as indicated in Section (Ablation Study of the Applied Filters) below, where some other filters have already been evaluated as well. Furthermore, the GFB filter achieves the desired result on the dataset. To feed the model with a better image and obtain a satisfactory result, we performed a variety of preprocessing approaches to the dataset, including image denoising, image annotation removes, image enhancement using CLAHE, and image filtering. Figure 7 depicts the final result after using all of the picture preparation techniques described.

## Ablation Study of the Applied Filters

A number of preprocessing techniques were applied to the dataset to obtain a satisfying outcome. The validation accuracy is given as 'Val Acc' in Table 3, whereas the test accuracy is indicated as 'T\_Acc'; validation loss is indicated as 'Val\_Loss' and test loss is indicated as 'T\_Loss'. From Table 3, fifteen filters have been evaluated to feed the model and obtain the highest accuracy. 'Green Fire Blue' achieved the highest of all the filters with Val\_Acc 98.03%, along with T\_Acc 98.89%, Val\_loss 0.11%, and T\_Loss 0.09%, respectively.

#### 3.2. Image Augmentation

To perform well, a deep learning algorithm requires a large number of inputs. In this study, many augmentation techniques are used to increase the enhanced data. By augmenting the training datasets with extra and unique samples, data augmentation may help to enhance the performance and results of the machine learning algorithms. If the datasets used to train the algorithm are suitably large and diverse, the method works better and is highly accurate. Through the utilization of image augmentation techniques, the accuracy of the findings is improved. Additionally, data augmentation techniques are an excellent tool for diversifying the datasets. Generally, data augmentation techniques have been used to expand the volume of the training sets in order to provide more relevant training data to large-capacity learners. However, there is a new tendency developing among the research community on deep learning in which samples are enhanced through the test data augmentation method [69–72]. Test data augmentation can help increase the robustness of a trained model [73–75]. Test data augmentation may be utilized to enhance deep network prediction performance and open up new intriguing possibilities for medical image interpretation [76–78]. The most often utilized methods for augmenting data include rotating, mirroring, flipping, zooming, and cropping.



Figure 7. The final outcome following the use of all image preprocessing methods.

Filters	Val_Acc (%)	Val_Loss (%)	T_Acc (%)	T_Loss (%)	Finding
5 Ramps	63.88	0.89	64.32	0.87	Accuracy Dropped
6 Shades	62.71	0.94	63.45	0.90	Accuracy Dropped
3-3-2 RGB	65.33	0.86	66.01	0.83	Accuracy Dropped
Blue Orange icb	72.59	0.72	73.77	0.68	Accuracy Dropped
16 colors	82.63	0.61	83.09	0.59	Accuracy Dropped
Spectrum	81.19	0.65	81.80	0.61	Accuracy Dropped
Sepia	85.40	0.49	86.90	0.39	Accuracy Dropped
Cyan Hot	85.67	0.45	86.11	0.41	Accuracy Dropped
Edges	83.55	0.57	83.99	0.53	Accuracy Dropped
Fire	87.92	0.37	87.47	0.40	Accuracy Dropped
Green Fire Blue	97.03	0.11	98.89	0.09	Identical Accuracy
Phase	88.76	0.25	88.23	0.29	Accuracy Dropped
Jet	83.52	0.58	83.85	0.54	Accuracy Dropped
Thallium	88.19	0.28	88.99	0.22	Accuracy Dropped
Viridis	95.10	0.19	95.50	0.16	Accuracy Dropped

Table 3. Analyses of performances of various filters.

In this study, the dataset is collected and merged from various sources as the amount of data in each class is not balanced (see Table 2). To balance the dataset, oversampling and undersampling methods are utilized. Firstly, a random undersampling strategy is applied to balance the class (control, COVID-19, Effusion, and Pneumonia) with excessive data. This method randomly deletes data from the majority classes and the number of data is reduced to 8000 data per class. Then, the oversampling (data augmentation) strategy is used to balance the class (Lung Opacity, Mass, Nodule, Pulmonary Fibrosis, Pneumothorax, and Tuberculosis) with insufficient data. Eight augmentation strategies are used in this study: Rotate 90° left, Rotate 90° right, Horizontal flip, Rotate 45° Horizontal, Rotate 45° Vertical, Rotate 45°, Translate and Vertical flip on pre-processed data. Table 4 summarizes all of the augmentation parameters. After augmentation, the dataset was increased to comprise 80,000 images (Control: 8000, COVID-19: 8000, Effusion: 8000, Lung Opacity: 8000, Mass: 8000, Nodule: 8000, Pulmonary Fibrosis: 8000, Pneumonia: 8000, Pneumothorax: 8000, Tuberculosis: 8000). Figure 8 shows the outcome of the data augmentation.

Table 4. Value of data augmentation variables.

Augmentation Parameters	Value	
Rotate left	90°	
Rotate right	90°	
Horizontal flip	True	
Rotate	$45^{\circ}$	
Rotate Horizontal	$45^{\circ}$	
Rotate Vertical	$45^{\circ}$	
Vertical flip	True	
Translate	x, y (29.0, 14.0)	

Before training begins, the dataset must be divided. The X-ray images were divided into three groups according to the 60:20:20 split across the training, validation, and test sets. After dividing the 80,000 X-ray images of the dataset into three subsets, the training set comprises 48,000 X-ray images, the validation set includes 16,000 X-ray images, and the testing set contains 16,000 X-ray images. Table 5 summarizes the final description of the dataset after all the pre-processing steps.



**Figure 8.** The outcome after balancing all class and using eight augmentation procedures: Rotate 90° left, Rotate 90° right, Horizontal flip, Rotate 45° Horizontal, Rotate 45° Vertical, Rotate 45°, Translate, Vertical flip.

Table 5. The final dataset after pre-processing.

Features	Properties
Number of Images	80,000
Number of Classes	10
Enhancement and Color Grading	CLAHE, Green Fire Blue
Number of Augmentation Techniques	8
Control	8000
COVID-19	8000
Effusion	8000
Lung Opacity	8000
Mass	8000
Nodule	8000
Pulmonary Fibrosis	8000
Pneumonia	8000
Pneumothorax	8000
Tuberculosis	8000

## 4. Proposed Model

One of the primary objectives of this study is to produce adequate classification results utilizing the transfer learning models on the merged dataset. Eight pre-trained models, including AlexNet, GoogLeNet, InceptionV3, ResNet50, MobileNetV2, EfficientNetB7, DenseNet121, and VGG16, are evaluated to determine the most efficient transfer learning method for the classification assignment. By adding several layers to the VGG16 model, a customized fine-tuned transfer learning method was developed and deployed to achieve the highest accuracy among all the other pre-trained models.

## 4.1. Transfer Learning Models

## 4.1.1. AlexNet

AlexNet is a popular convolutional neural network model. Max pooling, convolutions and dense layers are among the key building components. The model is fitted using grouping convolutions over two GPUs. Alexnet is made of eight layers within each set of learnable parameters. The model consists of five convolutional layers that combine fully connected and max pooling layers, along with two normalizing layers and one softmax layer. Each layer is built of a convolutional layer and an ReLU-based nonlinear activation function. Max pooling is accomplished by the usage of pooling layers. The input size  $224 \times 224 \times 3$  pixel is fixed due to the perceived layers. An RGB image is created whenever a gray picture is utilized as an input, and this is done by multiplying the single channel. The batch size is 128, and the model includes 60 million parameters.

#### 4.1.2. GoogLeNet

GoogLeNet is a convolutional neural network architecture modeled after the inception architecture. Googlenet attempts to increase computing efficiency by using an iterative algorithm as the fundamental layer. Subsequent layers are added on top of each other and perform parallel filtering on the input from the preceding layer. It makes use of inception modules, which let the network choose from a range of convolutional filter sizes inside each block. With a stride of two, these modules may be layered on top of each other to create an inception network that reduces the grid's resolution. In comparison to the inception design, GoogLeNet is a deep neural model with 22 layers and fewer parameters. The network has been pre-trained to accept pictures with a quality of  $224 \times 224$  pixel. The design is composed of three layers: Activation, AveragePooling2D, and Dense. Global average pooling was used in GoogLeNet rather than a fully connected layer.

### 4.1.3. InceptionV3

InceptionV3 reduces the amount of computer power required by updating prior inception designs. Factorized convolutions, normalization, down sampling, and parallelized calculations are only a few of the techniques used to minimize the computational cost. The training time of the InceptionV3 model is decreased by replacing smaller convolutions with larger convolutions. Numerous optimization techniques for removing restrictions and increasing the flexibility of an InceptionV3 model have been developed. Data preprocessing is a vital element as it has a considerable influence on the maximum accuracy of the model during training. At a minimum, pictures should be classified and scaled to correspond to the model. Pictures must be  $299 \times 299 \times 3$  pixel in size for Inception. Average pooling, convolutions, max pooling, dropouts, concatenations, and fully-connected layers are used to create the model. The model makes extensive use of batch normalization on activation inputs. The loss is calculated using SoftMax.

#### 4.1.4. ResNet50

ResNet50 is a 50-layer neural network (CNN) composed of 48 fully connected layers, a max pool layer, and an average pool layer. It is capable of doing floating-point calculations upwards of  $3.8 \times 10^9$ . The ResNet50 architecture makes use of a mixture of convolutional filters of varying sizes to overcome the decay issue inherent in CNN models and to shorten the time for training connected with the deep structure. ResNets contain few filters and so perform more quickly. The performance of the 34-layer ResNet is 3.6 billion FLOPS, compared with 1.8 billion FLOPS for the shorter 18-layer ResNets. Around 23 million variables may be used to train this design. The network may accept an input image with dimensions that are multiples of 32 in width, height, and channel width. Each ResNet architecture employs a  $7 \times 7$  kernel size for early convolution and a  $3 \times 3$  kernel size for max pooling, respectively. Each two-layer block in the 34-layer ResNet.

### 4.1.5. MobileNetV2

In MobileNetV2, a unique module which includes an inverting residual structure is added. MobileNetV2 enables the identification and segmentation of objects at the cutting-edge of technology. The design of MobileNetV2 starts with a fully connected convolution layer composed of 32 filters and 19 residual bottlenecks layers. MobileNetV2

is a convolutional neural network with 53 layers. It is a pre-trained classifier that has been trained on over a million photos from ImageNet. The network typically needs 3 billion multiply-add processes and makes use of 3.4 million variables. The model was pre-trained on photos classified into 100 distinct item categories. As a result, the network has retrieved a large number of characteristics from a diverse group of photos. The topology of MobileNetV2 starts with such fully convolutional layers comprised of 32 filters and 19 residual bottlenecks. It is divided into two different blocks, each with three layers. Both blocks start and conclude with a  $1 \times 1$  convolution layer containing 32 filters, whereas the second level layer is indeed a depth-wise fully connected layer. Throughout various levels of the architecture, the ReLU is employed. The difference between the two blocks lies within the stride size, having blocks 1 stride length of 1, while block two has a stride length of 2.

#### 4.1.6. EfficientNetB7

A new benchmark network was developed utilizing the AutoML MNAS architecture to boost performance, which increases the overall effectiveness and precision (FLOPS). In that it uses mobile inversion bottleneck convolution (MBConv), the resulting architecture is similar to MobileNetV2 and MnasNet; however, it is somewhat larger due to an increased FLOP limit. FLOPS are used to evaluate how well the algorithm performs in terms of correctness and efficiency. The MBConv is used in this architecture. The researchers subsequently increased the size of this first network in order to create the EfficientNets series of deep learning models. EfficientNets are a series of models derived from the original EfficientNet. The pre-trained weights are not included in EfficientNetB7. EfficientNet-B7, the most recent version of EfficientNet, offers the best accuracy of all of the versions and requires the fewest parameters.

#### 4.1.7. DenseNet121

In a feed-forward CNN, each convolutional layer saves the first, obtains the output from the preceding convolutional layer and generates outputs from extracted features which are then carried onto the next convolutional layer. As a result, there seem to be 'L' direct connections between 'L' layers, one between every layer and the following layer. Nevertheless, as the number of hidden layers inside the CNN increases, the 'vanishing gradient' issue becomes apparent. This implies that when the route from the feed to the production layers becomes longer, some information may 'vanish' or get lost, impairing the network's capacity to train properly. DenseNets address this issue by altering the typical CNN design and streamlining the layer connection pattern. Each layer in a DenseNet design is directly linked to every other layer, resulting in the term Densely Linked Convolutional Network. Between 'L'-levels, there are L(1+1)/2 direct links. No average is used; instead, the extracted features from previous levels are combined and used as inputs in the subsequent layers. By rejecting duplicate feature maps, DenseNets are able to reuse features as they need less parameters than a similar traditional CNN. In dense blocks, the quantity of filters fluctuates; however, the size of the feature maps does not. These intermediate layers, known as transition layers, are accountable for down-sampling the picture utilizing batch normalization,  $1 \times 1$  convolution, and  $2 \times 2$  pooling layers.

## 4.1.8. VGG16

The VGG16 DCNN model was developed by Simonyan and Zisserman [79]. Increasing the depth of the VGG model may help the kernels in learning more complex properties. A pre-trained VGG16 network beat the fully trained networks by a large margin in experiments on the effectiveness of transfer learning [80]. This network is trained on images from the ImageNet collection totaling over a million. The network is composed of 16 layers and is capable of recognizing images belonging to a variety of categories. The conv1 layer is fed  $224 \times 224$  RGB pictures of standard size. The picture is convoluted using a succession of filters tuned to create the lowest possible receptive field:  $3 \times 3$ . Furthermore, one of the

pairings utilizes a  $1 \times 1$  convolutional filter, which changes the input channels linearly. The stride of convolution is set to one pixel, and the spatial padding of the convolution layers' input is adjusted to retain spatial resolution after convolution. Padding, for example, is 1 pixel for  $3 \times 3$  convolutional layers. Following the convolutional layers, five max-pooling layers are employed to achieve spatial pooling. Stride 2 is being utilized to perform maxpooling operations across  $2 \times 2$  pixel frames. Three convolutional layers are placed on top of one another. The first two contain 4096 channels, respectively, whereas the third does ILSVRC classification on a 1000-way basis. Finally, there is a softmax layer. In all networks, the structure of the entire linked levels is the same.

## 4.1.9. LungNet22

In terms of classification performance, the fine-tuned VGG16 architecture named LungNet22 surpasses the other eight model architectures discussed above. As a consequence, the LungNet22 model is proposed and validated using our merged dataset, which is constructed on VGG16 architecture. Furthermore, ablation research is being carried out to improve the architecture's endurance concerning lung disease classification. The architecture of the model is illustrated in Figure 9.



Figure 9. The architecture of LungNet22, a fine-tuned VGG16 network.

The dataset contains 80,000 images classified into ten categories: Control, COVID-19, Effusion, Lung Opacity, Mass, Nodule, Pulmonary Fibrosis, Pneumonia, Pneumothorax and Tuberculosis. The dataset is divided into 60% for training, 20% for validation, and 20% for testing. All tests were conducted on an AMD Ryzen 7 3800X, with base clock 3.90–4.50 GHz CPU with eight (32 MB L3 cache, storage speed = 3200 MHz) cores and sixteen threads, 64 GB of RAM, and the Anaconda 3 (Jupyter Notebook) platform. It is integrated with an AMD Radeon RX 580 series GPU. The Python programming language is used to obtain the findings.

LungNet22 is formed by connecting two blocks after VGG16's fifth block. The sixth block has three convolutional layers and one GlobalAveragePooling2D layer, while the seventh block contains a flatten layer connected to a dense layer. As the input layer of the network default required a size of  $224 \times 224 \times 3$  pixel for the picture, it is an RGB image. The first block in this proposed architecture has two convolutional layers with 64 channels of  $3 \times 3$  kernel size, an ReLU activation function, and the same padding preceded by a Maxpooling2D layer of  $2 \times 2$  stride and a pool size of  $2 \times 2$ . Similar to the first block, the second block has two convolution layers with 128 channels of kernel size  $3 \times 3$ , an ReLU activation function and the same padding preceded by a Maxpooling2D layer with a stride of  $2 \times 2$  and a pool size of  $2 \times 2$ . In blocks 3, 4, and 5 are three convolutional layers followed by one Maxpooling2D layer. These three convolutional layers have channel lengths of 256, 512, and 512, sequentially, with the same kernel size of  $3 \times 3$ , the same padding, and an ReLU activation function. All of these blocks with a Maxpooling2D layer also performed as a last layer with a stride and a pool size of  $2 \times 2$ . Each Maxpooling2D layer compresses the input image to twice its initial value size.

Block six has three convolutional layers preceded by a GlobalAveragePooling2D layer. The convolutional layers have the same kernel size of  $3 \times 3$ , the same padding, and an ReLU activation function with 512 channels. A flatten layer has been also added in block 7, connected to dense a layer which outputs for the 10 classes. The Softmax activation and Adam optimizer were applied to the final layer. A speed of learning of 0.000001 is used throughout the method. Finally, we assessed the accuracy, recall, precision, and f1-score.

### 4.2. Ablation Study of the Proposed Model (LungNet22)

Four experiments were conducted as an ablation study by modifying various aspects of the proposed LungNet22 architecture depending upon the fine-tuned VGG16 model. By modifying different components, it is possible to develop a more resilient architecture with increased classification accuracy. The following aspects were subjected to an ablation study: Global AveragePooling2D (GAP2D), Flatten layer, Loss function, Optimizer, and Learning rate.

## 4.2.1. Ablation Study 1: Changing GlobalAveragePooling2D Layer

In this study, the validation loss is indicated by the abbreviation 'Val\_Loss'; the validation accuracy is represented by the abbreviation 'Val\_Acc'; the test loss is defined by the abbreviation 'T\_Loss'; and the test accuracy is stated by the abbreviation 'T\_Acc'. Table 6 shows that for the GlobalAveragePooling2D layer, the LungNet22 has the greatest results, with a Val\_Acc of 97.03% and a T\_Acc of 98.89%. However, the accuracy of the GlobalMaxPooling2D, AveragePooling2D, and Maxpooling2D layers decreases slightly, with Val\_Acc values of 95.73%, 94.92%, and 95.77%, and T\_Acc values of 96.12%, 95.30%, and 96.19%, consecutively.

Table 6. Changing the GlobalAveragePooling2D (GAP2D) layer to evaluate the ablation study.

Case Study	Layer Name	Val_Acc (%)	Val_Loss (%)	T_Acc (%)	T_Loss (%)	Finding
	GlobalAveragePooling2D	97.03	0.11	98.89	0.09	Identical Performance
1	GlobalMaxPooling2D	95.73	0.27	96.12	0.23	Accuracy Dropped
1	AveragePooling2D	94.92	0.35	95.30	0.27	Accuracy Dropped
	MaxPooling2D	95.77	0.21	96.19	0.25	Accuracy Dropped

#### 4.2.2. Ablation Study 2: Changing the Flatten layer

The validation accuracy is labeled 'Val\_Acc'; the validation loss is labeled 'Val\_Loss'; the test accuracy is labeled 'T\_Acc'; and the test loss is labeled 'T\_Loss'. GlobalAveragePooling2D, GlobalMaxPooling2D, AveragePooling2D, and Maxpooling2D change the flatten layer to see how it affects the model effectiveness. As shown in Table 7, GlobalAveragePooling2D's accuracy drops marginally with the T\_Acc of 96.94%. The network achieves

the lowest results for GlobalMaxPooling2D, AveragePooling2D, and Maxpooling2D, respectively, with T\_Acc values of 91.12%, 93.21%, and 95.16%.

Case Study	Layer Name	Val_Acc (%)	Val_Loss (%)	T_Acc (%)	T_Loss (%)	Finding
	Flatten	97.03	0.11	98.89	0.09	Identical Performance
	GlobalAveragePooling2D	96.12	0.25	96.94	0.22	Accuracy Dropped
2	GlobalMaxPooling2D	90.73	0.67	91.12	0.58	Accuracy Dropped
	AveragePooling2D	92.95	0.53	93.21	0.47	Accuracy Dropped
	MaxPooling2D	94.78	0.40	95.16	0.36	Accuracy Dropped

Table 7. Changing the Flatten layer to evaluate the ablation study.

4.2.3. Ablation Study 3: Changing Loss Functions

To identify the ideal loss function for the new proposal, a variety of loss functions were explored, namely Cosine Similarity, Categorical Crossentropy, and Mean Squared Error. Table 8 illustrates the model's performance using the specified loss functions. When fitted with Categorical Crossentropy, the model achieved the maximum test accuracy of 98.89 percent. The loss functions for Cosine Similarity and Mean Squared Error resulted in a modest decrease in test accuracy of 96.16% and 94.20%, respectively. To attain the greatest classification performance, the Categorical Crossentropy loss function was selected for further exploration.

Table 8. Changing the Loss Function to evaluate the ablation study.

Case Study	Loss Function	Val_Acc (%)	Val_Loss (%)	T_Acc (%)	T_Loss (%)	Finding
	Categorical Crossentropy	97.03	0.11	98.89	0.09	Identical Performance
3	Mean Squared Error	93.77	0.67	94.20	0.58	Accuracy Dropped
	Cosine Similarity	95.82	0.41	96.19	0.37	Accuracy Dropped

4.2.4. Ablation Study 4: Changing Optimizer and Learning Rate

The validation loss is labeled 'Val\_Loss'; the validation accuracy is labeled 'Val\_Acc'; the test loss is labeled 'T\_Loss'; and the test accuracy is labeled 'T\_Acc'. As shown in Table 9, the optimizer 'Adam' with a learning rate of 0.000001 increased the model performance with the T\_Acc of 98.89% and the Val\_Acc of 97.03%. The 'Adam' optimizer still had the least T\_Loss of 0.09, with a learning rate of 0.000001. Other optimizers, such as SGD, RMSprop, and Nadam, scored very efficiently in our model, with T Acc values above 90%.

Table 9. Changing Optimizer and Learning rate to evaluate the ablation study.

Case Study	Optimizer	Learning Rate	Val_Acc (%)	Val_Loss (%)	T_Acc (%)	T_Loss (%)	Finding
4		0.000001	97.03	0.11	98.89	0.09	Identical Performance
	Adam	0.0001	94.50	0.64	95.07	0.47	Accuracy dropped
		0.00001	95.61	0.46	96.22	0.36	Accuracy dropped
		0.001	93.35	0.69	93.98	0.65	Accuracy dropped
		0.01	91.96	0.77	92.35	0.71	Accuracy dropped

Case Study	Optimizer	Learning Rate	Val_Acc (%)	Val_Loss (%)	T_Acc (%)	T_Loss (%)	Finding
		0.000001	95.12	0.27	95.88	0.21	Accuracy dropped
		0.0001	94.83	0.65	95.15	0.60	Accuracy dropped
	SGD	0.00001	94.97	0.63	95.70	0.58	Accuracy dropped
		0.001	90.65	0.80	92.08	0.75	Accuracy dropped
		0.01	92.57	0.72	93.75	0.64	Accuracy dropped
		0.000001	94.57	0.64	94.90	0.49	Accuracy dropped
		0.0001	92.27	0.69	93.20	0.57	Accuracy dropped
	RMSprop	0.00001	93.06	0.58	94.51	0.68	Accuracy dropped
		0.001	92.13	0.72	93.90	0.51	Accuracy dropped
		0.01	91.74	0.80	92.56	0.62	Accuracy dropped
		0.000001	95.06	0.25	95.89	0.23	Accuracy dropped
		0.0001	94.26	0.75	95.11	0.68	Accuracy dropped
	Nadam	0.00001	94.76	0.70	95.32	0.65	Accuracy dropped
		0.001	93.44	0.72	94.81	0.70	Accuracy dropped
		0.01	92.35	0.65	94.20	0.67	Accuracy dropped

#### Table 9. Cont.

#### 5. Results

The study used a total of 80,000 frontal chest X-ray image data that were divided into 10 classes, namely Control, COVID-19, Effusion, Lung Opacity, Mass, Nodule, Pulmonary Fibrosis, Pneumonia, Pneumothorax, and Tuberculosis. Each class contains 8000 image data. These data are preprocessed before CNN models are applied to them. All the images in the balanced dataset are resized to  $224 \times 224$  pixel to ensure the same scale. The dataset is divided into training, validation, and test sets. The training set is comprised of 60% or 48,000 data. The validation and test sets are both made up of 20% data from that dataset, i.e., 16,000 data in each set. Initially, eight pre-trained CNN models are trained, validated and tested. From them, the model with most accurate performance is selected for fine-tuning to create a novel LungNet CNN model to classify lung disease.

To begin with, eight pre-trained CNN models are trained and tested on the dataset. The models are AlexNet, GoogLeNet, InceptionV3, ResNet50, MobileNetV2, Efficient-NetB7, DenseNet121, and VGG16. Each model performance has 300 epochs to classify the 10 classes. As shown in Figure 10a, VGG16 classifies the data with 92.95% accuracy, followed by MobileNetV2 with 90.90% accuracy. InceptionV3 classifies data with 88.98% accuracy. Subsequently, EfficientNetB7, GoogLeNet, DenseNet121, AlexNet, and ResNet50 can classify the data with 87.85%, 81.98%, 80.77%, 78.09%, and 75.85% accuracy, respectively. From Figure 10b, it is observed that the highest loss is shown by GoogLeNet and the lowest loss by MobileNetV2, with 0.61% and 0.29%, respectively. The model VGG16 has a loss of 0.36%.

It is observed that the VGG16 has the highest classification accuracy among the implemented pre-trained models. Fine-tuning the said model, the proposed LungNet22 is established. The proposed model performs 300 epochs on the dataset. To optimize loss, optimizer 'Adam' is employed. The learning rate of the model is 0.000001. The model is fed with pre-processed image of  $224 \times 224$  pixels to achieve optimal accuracy.



**Figure 10.** The eight pre-trained models: (**a**) Accuracy and (**b**) Loss comparative graph to classify lung disease.

Figure 11a shows the training and test accuracy of the proposed model over the 300 epochs. It can be observed that the accuracy curve overall has increased significantly and that with the increasing epoch denotation the performance of the model improves considerably. In the first epoch, the training accuracy of the model is 53.08% and over the epochs increased to 96.47% in the last epoch. Similarly, the test accuracy of the first epoch is 44.6%, which increased to 98.89% in the last epoch. On the other hand, in Figure 11b the loss curve of the model is presented. Unlike the accuracy curve, the loss value of the model decreased with each epoch, denoting that the performance of the model increased. Initially, the training loss was 55.56%, and the test loss was 44.6%. However, on the final epoch, the training and test loss reduced significantly to 0.15% and 0.09%, respectively.



**Figure 11.** The (**a**) Accuracy and (**b**) Loss of the proposed model LungNet22 for 300 epochs using 'Adam' optimizer with learning rate 0.000001.

The proposed LungNet22 can successfully classify lung disease with high accuracy. The test set contains a total of 16,000 image data. Each of the 10 classes contains 1600 data. In Figure 12, the confusion matrix of the model is illustrated. According to the confusion matrix, the model can classify a total of 15,823 data correctly. Of these, 1597 belong to the Control class and 1584 belong to the COVID-19 class. The model can accurately classify 1573, 1594, and 1560 data from the Effusion, Lung Opacity, and Mass classes, respectively. From the classes of Nodule, Pulmonary Fibrosis, Pneumonia, Pneumothorax, and Tuberculosis,

	Tuberculosis -	0	0	1	1	0	2	0	0	0	1596
	Pneumothorax -	0	1	0	1	0	2	0	4	1592	0
	Pneumonia -	5	7	0	7	2	9	0	1561	9	0
	Pulmonary Fibrosis -	0	5	0	3	2	0	1580	1	0	9
Truth	Nodule -	2	0	5	0	7	1586	0	0	0	0
	Mass -	10	9	0	2	1560	0	5	6	0	8
	Lung Opacity -	0	0	2	1594	0	1	3	0	0	0
	Effusion -	4	5	1573	3	7	0	6	0	1	1
	COVID-19 -	3	1584	0	2	0	4	1	1	5	0
	Control -	1597	0	0	0	2	0	1	0	0	0
		Control -	COVID-19 -	Effusion -	Lung Opacity -	Mass -	Nodule -	Pulmonary Fibrosis -	Pneumonia -	Pneumothorax -	Tuberculosis -
						Pre	dicted				

the model can correctly identify 1586, 1580, 1561, 1592, and 1596 data, respectively. A total of 177 data are misclassified by the proposed LungNet22 model.

Figure 12. Confusion matrix for LungNet22 model.

From the confusion matrix in Figure 12, the performance of Lungnet22 is calculated. To determine the performance, measures such as accuracy, precision, recall, specificity, and f1-score are considered where accuracy refers to the percentage of corrected classified data. Precision refers to the percentage of positive classified data that was relevant. Recall or sensitivity indicates the percentage of correctly predicted positive data from all the positive data. Similarly, Specificity denotes the percentage of correctly predicted negative data from all the negative data. The F1-score expresses the balance between the precision and the recall values. The mathematical expression of the measures is shown in Equations (14)–(18).

$$Accuracy = \frac{TP + TN}{TP + TN + FP + FN}$$
(14)

$$Precision = \frac{TP}{TP + FP}$$
(15)

$$Sensitivity/Recall = \frac{TP}{TP + FN}$$
(16)

Specificity = 
$$\frac{TN}{TN + FP}$$
 (17)

$$F1 - Score = 2\left(\frac{Precision \times Recall}{Precision + Recall}\right)$$
(18)

Here, TP or true positive is the number of correctly identified positive data in a class. TN or true negative refers to the number of correctly identified negative data. FP, also known as false positive, is the number of negative data identified as positive. Similarly, FN or false negative is the number of positive data identified as positive.

The performance of the proposed model is analyzed for each class of the dataset. In Figure 13, the performance measures are shown. It is observed that the model performance is high for each class. The accuracy and specificity across all the classes remained consider-

ably consistent at over 99.60% for both. The lowest precision of the model is observed for the COVID-19 class at 98.32%, and the highest precision is for the Effusion class at 99.49%. Similarly, the Mass class shows the lowest recall value at 97.50%, and the Control class has the highest value at 99.81%. Finally, the highest F1-score is achieved in the Tuberculosis class, and the lowest is in Mass class at 99.32% and 98.11%, respectively.



Figure 13. Performance score of LungNet22 network.

A number of research attempts have been performed to improve the practicability and explicability of deep learning. Additionally, it is vital to enhance the understandability of deep neural models in various deep learning applications for medical imaging. Selvaraju et al. [81] demonstrated the functioning of deep learning using a method entitled Gradient Weighted Class Activation Mapping (Grad-CAM). Grad-CAM visualizes any densely linked neural network. This enables the model's extra information to be determined while performing classification or prediction operations. The input is a standard X-ray image, and the proposed framework is employed as a detection approach. Grad-CAM is performed to the final convolution layer just after the proposed model predicts the label. Figure 14 depicts the heatmaps on the X-ray pictures that are visualized using the proposed methodology.

For performance evaluation of the proposed LungNet22, an ROC (Receiver Operating Characteristics) curve is constructed for each class. The "one class versus rest" method is applied for the construction of the ROC curve for a multiclass classification task. An ROC curve is essentially a Sensitivity vs. 1-Specificity curve, where the values range between 0 to 1, with 1 indicating the best performance and 0 indicating an unsatisfactory performance. In Figure 15, the ROC curve of each class is presented. From the ROC curve, the AUC (Area Under Curve) value is computed. The AUC value of each class is recorded. The AUC value of the Control class is 0.9861. For the COVID-19, Effusion, Lung Opacity, Mass, Nodule, Pulmonary Fibrosis, Pneumonia, Pneumothorax, and Tuberculosis classes, the AUC value is 0.9745, 0.9842, 0.9813, 0.9798, 0.9773, 0.9699, 0.9721, 0.9655, and 0.9895 respectively. The AUC values overall are close to 1 for the 10 classes in the dataset, indicating that the model performs very efficiently in the classification of the lung images in each class.



**Figure 14.** Representation of infected area of lung in X-ray images utilizing Grad-CAM on the LungNet22 model.



Figure 15. ROC curve for each class of LungsNet22.

## 6. Discussion

The research experiments evaluated the classification and detection of lung diseases, as well as the diagnosis of various lung diseases (COVID-19, Effusion, Lung Opacity, Mass, Nodule, Pulmonary Fibrosis, Pneumonia, Pneumothorax, and Tuberculosis) using eight pre-trained deep CNN models. On the base of a fine-tuned VGG16 model, the LungNet22 model is presented. To determine the best model, several training/validation ratios were utilized. Aside from accuracy and AUC, the most often utilized performance measures for lung disease detection are recall, specificity, and the F1 score, which have all been used to perform an overall study on the predictive accuracy and durability of the models. On our merged X-ray dataset, the proposed LungNet22 model obtained the highest accuracy of

98.89%, the highest specificity of 98.893%, the highest F1 score of 0.988, and the highest recall of 98.893%. Not only do our findings demonstrate a considerable increase in AUC, but they also give a more comprehensive estimate measure of accuracy, F1 score, specificity, and recall.

## 7. Conclusions

Lung illnesses are widespread worldwide. Lung Opacity, COVID-19, Effusion, Mass, Pulmonary Fibrosis, Nodule, Pneumothorax, Pneumonia, and Tuberculosis are a few examples of these. It is important to diagnose lung illness quickly. For this objective, several image processing and artificial intelligence models have been formed. The purpose of this work is to propose a fine-tuned model called LungNet22 for classifying lung diseases such as COVID-19, Lung Opacity, Mass, Effusion, Pulmonary Fibrosis, Pneumonia, Nodule, Pneumothorax, and Tuberculosis along with the Control class. The method of image annotation removal was utilized to erase the annotations from the images. Additionally, image enhancement methods were used to increase the quality of the X-ray images. The pre-processed image dataset was balanced and augmented to 80,000 images using eight augmentation techniques to increase the volume and balance of each X-ray dataset class. Eight pretrained classifiers and a fine-tuned model were evaluated to ensure the best level of accuracy. The structure of LungNet22 is based on the VGG16 model, but with new layers and hyper-parameters. The hypothesized network was conducted in an ablation study to determine how well it performed under different hyper-parameter conditions. LungNet22 scored best while using the Adam optimizer and a learning rate of 0.000001, with 96.47% training accuracy, 97.03% validation accuracy, and 98.89% testing accuracy. While performing recognition or prediction operations, the Grad-CAM was used to help in determining extra information about our proposed model LungNet22. The model with image processing, fine-tuning, and the ablation experiment has a high rate of accurate identification. The outcomes of this research demonstrate that image processing and enhancing and balancing techniques may help improve the model's performance. As a future study, our model could be used in the categorization of other diseases using CTscan pictures, or a hybrid model could be presented to classify the many types of lung illnesses. Cross-validation may be used to validate the assessment result and to provide mean accuracy and standard deviation on the training set as well as accuracy on the test set.

In the supplementary materials there is an instruction video [S1] showing how to run the code and a bit on the original dataset (the complete runnable code is in the supplementary material file S1, sample dataset is provided in the supplementary material file S2).

**Supplementary Materials:** The following supporting information can be downloaded at: https://www.mdpi.com/article/10.3390/jpm12050680/s1, Video S1: Instruction file on how to run the code. File S1: the complete runnable code. File S2: a sample of the dataset.

**Author Contributions:** Conceptualization, F.M.J.M.S. and S.A.; methodology, F.M.J.M.S., S.A. and A.K.; software, F.M.J.M.S., Z.T. and P.G.; validation, F.M.J.M.S., S.A. and R.I.; formal analysis, S.A. and A.K.; investigation, F.M.J.M.S. and R.I.; resources, P.G., R.I. and A.K.; data curation, F.M.J.M.S. and R.I.; writing—original draft preparation, F.M.J.M.S., Z.T., S.A. and A.K.; writing—review and editing, S.A. and A.K.; visualization, F.M.J.M.S. and Z.T.; supervision, S.A. and F.D.B.; project administration, A.K. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** The datasets used for this study can be obtained from the sources mentioned in References [21–36]. The complete set of code can be also be obtained from the following URL: https://github.com/Shamrat777/Lung-Disease-Classiffication (accessed on 12 April 2022). The Supplementary Material section has the zip file (106 MB) containing the complete runnable code, a sample of the dataset, and a short video and instruction file on how to run the code. Because the

dataset provided is a sample of the original one, the result will be different. The sample contains 5000 images, while the original dataset had 80,000 and was over 12 GB in size.

**Conflicts of Interest:** The authors declare no conflict of interest.

# References

- Bhandary, A.; Prabhu, G.A.; Rajinikanth, V.; Thanaraj, K.P.; Satapathy, S.C.; Robbins, D.E.; Shasky, C.; Zhang, Y.D.; Tavares, J.M.; Raja, N.S. Deep-learning framework to detect lung abnormality–A study with chest X-ray and lung CT scan images. *Pattern Recognit. Lett.* 2020, 129, 271–278. [CrossRef]
- Han, T.; Nunes, V.X.; Souza, L.F.D.F.; Marques, A.G.; Silva, I.C.L.; Junior, M.A.A.F.; Sun, J.; Filho, P.P.R. Internet of Medical Things—Based on Deep Learning Techniques for Segmentation of Lung and Stroke Regions in CT Scans. *IEEE Access* 2020, *8*, 71117–71135. [CrossRef]
- 3. De Sousa, P.M.; Carneiro, P.C.; Oliveira, M.M.; Pereira, G.M.; Junior, C.A.D.C.; de Moura, L.V.; Mattjie, C.; da Silva, A.M.M.; Patrocinio, A.C. COVID-19 classification in X-ray chest images using a new convolutional neural network: CNN-COVID. *Res. Biomed. Eng.* **2021**, *38*, 87–97. [CrossRef]
- Tobias, R.R.; De Jesus, L.C.; Mital, M.E.; Lauguico, S.C.; Guillermo, M.A.; Sybingco, E.; Bandala, A.A.; Dadios, E.P. CNN-based deep learning model for chest X-ray health classification using tensorflow. In Proceedings of the 2020 RIVF International Conference on Computing and Communication Technologies (RIVF), Ho Chi Minh City, Vietnam, 14–15 October 2020; pp. 1–6.
- Zhou, L.; Yin, X.; Zhang, T.; Feng, Y.; Zhao, Y.; Jin, M.; Peng, M.; Xing, C.; Li, F.; Wang, Z.; et al. Detection and Semiquantitative Analysis of Cardiomegaly, Pneumothorax, and Pleural Effusion on Chest Radiographs. *Radiol. Artif. Intell.* 2021, *3*, e200172. [CrossRef] [PubMed]
- 6. Rajaraman, S.; Antani, S. Training deep learning algorithms with weakly labeled pneumonia chest X-ray data for COVID-19 detection. *medRxiv* **2020**. preprint.
- Akter, S.; Shamrat, F.M.; Chakraborty, S.; Karim, A.; Azam, S. COVID-19 detection using deep learning algorithm on chest X-ray images. *Biology* 2021, 10, 1174. [CrossRef] [PubMed]
- 8. Hashmi, M.F.; Katiyar, S.; Keskar, A.G.; Bokde, N.D.; Geem, Z.W. Efficient Pneumonia Detection in Chest Xray Images Using Deep Transfer Learning. *Diagnostics* 2020, *10*, 417. [CrossRef]
- Chatterjee, S.; Dzitac, S.; Sen, S.; Rohatinovici, N.C.; Dey, N.; Ashour, A.S.; Balas, V.E. Hybrid modified Cuckoo Search-Neural Network in chronic kidney disease classifi cation. In Proceedings of the 2017 14th International Conference on Engineering of Modern Electric Systems (EMES), Oradea, Romania, 1–2 June 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 164–167.
- Islam, K.; Wijewickrema, S.; Collins, A.; O'Leary, S. A deep transfer learning frame work for pneumonia detection from chest X-ray images. In Proceedings of the 15th International Joint Conference on Computer Vision, Imaging and Computer Graphics Theory and Applications, Valletta, Malta, 27–29 February 2020; SCITEPRESS—Science and Technology Publications: Setúbal, Portugal, 2020. [CrossRef]
- 11. Salehinejad, H.; Colak, E.; Dowdell, T.; Barfett, J.; Valaee, S. Synthesizing chest X-ray pathology for training deep convolutional neural networks. *IEEE Trans. Med. Imaging* **2018**, *38*, 1197–1206. [CrossRef]
- 12. Demir, F.; Sengur, A.; Bajaj, V. Convolutional neural networks based efficient approach for classification of lung diseases. *Health Inf. Sci. Syst.* **2019**, *8*, 4. [CrossRef]
- 13. Jin, B.; Cruz, L.; Goncalves, N. Deep Facial Diagnosis: Deep Transfer Learning from Face Recognition to Facial Diagnosis. *IEEE Access* 2020, *8*, 123649–123661. [CrossRef]
- 14. Rauf, H.T.; Saleem, B.A.; Lali, M.I.U.; Khan, M.A.; Sharif, M.; Bukhari, S.A.C. A citrus fruits and leaves dataset for detection and classification of citrus diseases through machine learning. *Data Brief* **2019**, *26*, 104340. [CrossRef] [PubMed]
- 15. Gao, J.; Wang, H.; Shen, H. Task Failure Prediction in Cloud Data Centers Using Deep Learning. *IEEE Trans. Serv. Comput.* **2020**. [CrossRef]
- Gao, J.; Wang, H.; Shen, H. Smartly handling renewable energy instability in supporting a cloud datacenter. In Proceedings of the 2020 IEEE International Parallel and Distributed Processing Symposium (IPDPS), New Orleans, LA, USA, 18–22 May 2020; IEEE: Piscataway, NJ, USA, 2020; pp. 769–778.
- 17. Jain, R.; Nagrath, P.; Kataria, G.; Kaushik, V.S.; Hemanth, D.J. Pneumonia detection in chest X-ray images using convolutional neural networks and transfer learning. *Measurement* 2020, *165*, 108046. [CrossRef]
- Mongan, J.; Moy, L.; Kahn, C.E. Checklist for Artificial Intelligence in Medical Imaging (CLAIM): A Guide for Authors and Reviewers. *Radiol. Artif. Intell.* 2020, 2, e200029. [CrossRef] [PubMed]
- 19. Ali, A.-R.; Li, J.; Yang, G.; O'Shea, S.J. A machine learning approach to automatic detection of irregularity in skin lesion border using dermoscopic images. *PeerJ Comput. Sci.* 2020, *6*, e268. [CrossRef] [PubMed]
- 20. Zhang, J.; Xie, Y.; Li, Y.; Shen, C.; Xia, Y. Covid-19 screening on chest x-ray images using deep learning based anomaly detection. *arXiv* 2020, arXiv:2003.12338.
- 21. Kaggle. Available online: https://www.kaggle.com/prashant268/chest-xray-covid19-pneumonia (accessed on 5 January 2022).
- 22. Kaggle. Available online: https://www.kaggle.com/sid321axn/covid-cxr-image-dataset-research (accessed on 5 January 2022).
- 23. Kaggle. Available online: https://www.kaggle.com/jtiptj/chest-xray-pneumoniacovid19tuberculosis (accessed on 5 January 2022).

- 24. Github. Available online: https://github.com/ieee8023/covid-chestxray-dataset (accessed on 3 January 2022).
- 25. Kaggle. Available online: https://www.kaggle.com/andyczhao/covidx-cxr2?select=train (accessed on 4 January 2022).
- Kaggle. Available online: https://www.kaggle.com/tawsifurrahman/tuberculosis-tb-chest-xray-dataset (accessed on 4 January 2022).
- Kaggle. Available online: https://www.kaggle.com/raddar/tuberculosis-chest-xrays-shenzhen?select=images (accessed on 5 January 2022).
- 28. Kaggle. Available online: https://www.kaggle.com/donjon00/covid19-detection (accessed on 5 January 2022).
- 29. Kaggle. Available online: https://www.kaggle.com/volodymyrgavrysh/pneumothorax-binary-classification-task (accessed on 5 January 2022).
- 30. Kaggle. Available online: https://www.kaggle.com/nih-chest-xrays/sample (accessed on 6 January 2022).
- 31. Kaggle. Available online: https://www.kaggle.com/kamildinleyici/covid-normal-viral-opacity-v2 (accessed on 3 January 2022).
- 32. Kaggle. Available online: https://www.kaggle.com/gauravduttakiit/x-ray-report (accessed on 4 January 2022).
- NIH Dataset. Available online: https://nihcc.app.box.com/v/ChestXray-NIHCC/folder/36938765345 (accessed on 3 January 2022).
- Kaggle. Available online: https://www.kaggle.com/homayoonkhadivi/chest-xray-worldwide-datasets (accessed on 3 January 2022).
- Kaggle. Available online: https://www.kaggle.com/tawsifurrahman/covid19-radiography-database (accessed on 6 January 2022).
- 36. Kaggle. Available online: https://www.kaggle.com/raddar/tuberculosis-chest-xrays-shenzhen (accessed on 6 January 2022).
- 37. Ibrahim, D.M.; Elshennawy, N.M.; Sarhan, A.M. Deep-chest: Multi-classification deep learning model for diagnosing COVID-19, pneumonia, and lung cancer chest diseases. *Comput. Biol. Med.* **2021**, *132*, 104348. [CrossRef]
- Bharati, S.; Podder, P.; Mondal, M.R.H. Hybrid deep learning for detecting lung diseases from X-ray images. *Inform. Med. Unlocked* 2020, 20, 100391. [CrossRef]
- Chen, J.I.Z. Design of accurate classification of COVID-19 disease in X-ray images using Deep Learning Approach. J. ISMAC 2021, 3, 132–148. [CrossRef]
- Sivasamy, J.; Subashini, T. Classification and predictions of lung diseases from chest X-rays using MobileNet. Int. J. Anal. Exp. Modal Anal. 2020, 12, 665–672.
- 41. Fauzan, A.R.; Wahyuddin, M.I.; Ningsih, S. Pleural Effusion Classification Based on Chest X-ray Images using Convolutional Neural Network. *J. Ilmu Komput. Inf.* **2021**, *14*, 9–16. [CrossRef]
- 42. Abbas, A.; Abdelsamea, M.M.; Gaber, M.M. Classification of COVID-19 in chest X-ray images using DeTraC deep convolutional neural network. *Appl. Intell.* **2021**, *51*, 854–864. [CrossRef] [PubMed]
- 43. Albahli, S.; Rauf, H.T.; Algosaibi, A.; Balas, V.E. AI-driven deep CNN approach for multi-label pathology classification using chest X-rays. *PeerJ Comput. Sci.* 2021, 7, e495. [CrossRef] [PubMed]
- 44. Podder, S.; Bhattacharjee, S.; Roy, A. An efficient method of detection of COVID-19 using Mask R-CNN on chest X-ray images. *AIMS Biophys.* **2021**, *8*, 281–290. [CrossRef]
- Singh, S.; Sapra, P.; Garg, A.; Vishwakarma, D.K. CNN based COVID-aid: COVID 19 Detection using Chest X-ray. In Proceedings of the 2021 5th International Conference on Computing Methodologies and Communication (ICCMC), Erode, India, 8–10 April 2021; pp. 1791–1797.
- 46. Moujahid, H.; Cherradi, B.; El Gannour, O.; Bahatti, L.; Terrada, O.; Hamida, S. Convolutional neural network based classification of patients with pneumonia using X-ray lung images. *Adv. Sci. Technol. Eng. Syst. J.* **2020**, *5*, 167–175. [CrossRef]
- 47. Das, A.K.; Ghosh, S.; Thunder, S.; Dutta, R.; Agarwal, S.; Chakrabarti, A. Automatic COVID-19 detection from X-ray images using ensemble learning with convolutional neural network. *Pattern Anal. Appl.* **2021**, *24*, 1111–1124. [CrossRef]
- 48. Irmak, E. COVID-19 disease severity assessment using CNN model. IET Image Proc. 2021, 15, 1814. [CrossRef]
- Hassantabar, S.; Ahmadi, M.; Sharifi, A. Diagnosis and detection of infected tissue of COVID-19 patients based on lung X-ray image using convolutional neural network approaches. *Chaos Solitons Fractals* 2020, 140, 110170. [CrossRef]
- Ismael, A.M.; Şengür, A. Deep learning approaches for COVID-19 detection based on chest X-ray images. *Expert Syst. Appl.* 2021, 164, 114054. [CrossRef]
- Wang, B.; Zhang, W. MARnet: Multi-scale adaptive residual neural network for chest X-ray images recognition of lung diseases. *Math. Biosci. Eng.* 2022, 19, 331–350. [CrossRef]
- 52. Denoising Autoencoders. Available online: https://omdena.com/blog/denoising-autoencoders/ (accessed on 15 January 2022).
- 53. Denoising Autoencoders. Available online: https://pyimagesearch.com/2020/02/24/denoising-autoencoders-with-kerastensorflow-and-deep-learning/ (accessed on 16 January 2022).
- 54. Liang, J.; Doermann, D.; Li, H. Camera-based analysis of text and documents: A survey. *Int. J. Doc. Anal. Recognit.* 2005, 7, 84–104. [CrossRef]
- 55. Zhang, S.; Liu, Y.; Jin, L.; Huang, Y.; Lai, S. Ensnet: Ensconce text in the wild. In Proceedings of the AAAI 2019 Conference on Artificial Intelligence, Honolulu, HI, USA, 27 January–1 February 2019; Volume 33, pp. 801–808.
- 56. Mirza, M.; Osindero, S. Conditional generative adversarial nets. arXiv 2014, arXiv:1411.1784.
- 57. Shen, X.; Chen, Y.C.; Tao, X.; Jia, J. Convolutional neural pyramid for image processing. *arXiv* 2017, arXiv:1704.02071.

- Dong, C.; Loy, C.C.; Tang, X. Accelerating the super-resolution convolutional neural network. In Proceedings of the 14th European Conference on Computer Vision, Amsterdam, The Netherlands, 8–16 October 2016; Springer: Berlin/Heidelberg, Germany; pp. 391–407.
- Yang, C.; Lu, X.; Lin, Z.; Shechtman, E.; Wang, O.; Li, H. High-resolution image inpainting using multi-scale neural patch synthesis. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition 2017, Honolulu, HI, USA, 21–26 July 2017; Volume 1, pp. 6721–6729.
- Johnson, J.; Alahi, A.; Li, F. Perceptual losses for real-time style transfer and super-resolution. In Proceedings of the European Conference on Computer Vision, Amsterdam, The Netherland, 8–16 October 2016; Springer: Berlin/Heidelberg, Germany, 2016; pp. 694–711.
- 61. Gatys, L.A.; Ecker, A.S.; Bethge, M. A neural algorithm of artistic style. arXiv 2015, arXiv:1508.06576. [CrossRef]
- 62. Gatys, L.A.; Ecker, A.S.; Bethge, M. Image style transfer using convolutional neural networks. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition 2016, Las Vegas, NV, USA, 27–30 June 2016; pp. 2414–2423.
- 63. Alshehri, A.; Taileb, M.; Alotaibi, R. DeepAIA: An Automatic Image Annotation Model based on Generative Adversarial Networks and Transfer Learning. *IEEE Access* 2022, *10*, 38437–38445. [CrossRef]
- Khvostikov, A.; Aderghal, K.; Benois-Pineau, J.; Krylov, A.; Catheline, G. 3D CNN-based classification using sMRI and MD-DTI images for Alzheimer disease studies. arXiv 2018, arXiv:1801.05968.
- 65. Mahmood, T.; Li, J.; Pei, Y.; Akhtar, F. An Automated In-Depth Feature Learning Algorithm for Breast Abnormality Prognosis and Robust Characterization from Mammography Images Using Deep Transfer Learning. *Biology* **2021**, *10*, 859. [CrossRef]
- 66. Wang, P.; Wang, J.; Li, Y.; Li, P.; Li, L.; Jiang, M. Automatic classification of breast cancer histopathological images based on deep feature fusion and enhanced routing. *Biomed. Signal. Process. Control.* **2021**, *65*, 102341. [CrossRef]
- 67. Zuiderveld, K. Contrast Limited Adaptive Histogram Equalization; ACM Digital Library: New York, NY, USA, 1994.
- 68. Hassan, N.; Ullah, S.; Bhatti, N.; Mahmood, H.; Zia, M. The Retinex based improved underwater image enhancement. *Multimed. Tools Appl.* **2021**, *80*, 1839–1857. [CrossRef]
- 69. Wang, G.; Li, W.; Aertsen, M.; Deprest, J.; Ourselin, S.; Vercauteren, T. Aleatoric uncertainty estimation with test-time augmentation for medical image segmentation with convolutional neural networks. *Neurocomputing* **2019**, *338*, 34–45. [CrossRef]
- Nalepa, J.; Marcinkiewicz, M.; Kawulok, M. Data Augmentation for Brain-Tumor Segmentation: A Review. Front. Comput. Neurosci. 2019, 13, 83. [CrossRef] [PubMed]
- Moshkov, N.; Mathe, B.; Kertesz-Farkas, A.; Hollandi, R.; Horvath, P. Test-time augmentation for deep learning-based cell segmentation on microscopy images. *Sci. Rep.* 2020, 10, 5068. [CrossRef] [PubMed]
- 72. Amiri, M.; Brooks, R.; Behboodi, B.; Rivaz, H. Two-stage ultrasound image segmentation using U-Net and test time augmentation. *Int. J. Comput. Assist. Radiol. Surg.* 2020, *15*, 981–988. [CrossRef]
- 73. Shanmugam, D.; Blalock, D.; Balakrishnan, G.; Guttag, J. Better aggregation in test-time augmentation. In Proceedings of the IEEE/CVF International Conference on Computer Vision, Montreal, BC, Canada, 11–17 October 2021; pp. 1214–1223.
- Jha, D.; Smedsrud, P.H.; Johansen, D.; de Lange, T.; Johansen, H.D.; Halvorsen, P.; Riegler, M.A. A Comprehensive Study on Colorectal Polyp Segmentation with ResUNet++, Conditional Random Field and Test-Time Augmentation. *IEEE J. Biomed. Health Inform.* 2021, 25, 2029–2040. [CrossRef] [PubMed]
- 75. Kandel, I.; Castelli, M. Improving convolutional neural networks performance for image classification using test time augmentation: A case study using MURA dataset. *Health Inf. Sci. Syst.* **2021**, *9*, 33. [CrossRef]
- 76. Hoar, D.; Lee, P.Q.; Guida, A.; Patterson, S.; Bowen, C.V.; Merrimen, J.; Wang, C.; Rendon, R.; Beyea, S.D.; Clarke, S.E. Combined Transfer Learning and Test-Time Augmentation Improves Convolutional Neural Network-Based Semantic Segmentation of Prostate Cancer from Multi-Parametric MR Images. *Comput. Methods Programs Biomed.* 2021, 210, 106375. [CrossRef]
- 77. Gonzalo-Martín, C.; García-Pedrero, A.; Lillo-Saavedra, M. Improving deep learning sorghum head detection through test time augmentation. *Comput. Electron. Agric.* 2021, 186, 106179. [CrossRef]
- Cohen, S.; Dagan, N.; Cohen-Inger, N.; Ofer, D.; Rokach, L. ICU Survival Prediction Incorporating Test-Time Augmentation to Improve the Accuracy of Ensemble-Based Models. *IEEE Access* 2021, 9, 91584–91592. [CrossRef]
- 79. Simonyan, K.; Zisserman, A. Very deep convolutional networks for large-scale image recognition. arXiv 2014, arXiv:1409.1556.
- Menon, L.T.; Laurensi, I.A.; Penna, M.C.; Oliveira, L.E.S.; Britto, A.S. Data Augmentation and Transfer Learning Applied to Charcoal Image Classification. In Proceedings of the 2019 International Conference on Systems, Signals and Image Processing (IWSSIP), Osijek, Croatia, 5–7 June 2019; pp. 69–74.
- Selvaraju, R.R.; Cogswell, M.; Das, A.; Vedantam, R.; Parikh, D.; Batra, D. Grad-cam: Visual explanations from deep networks via gradient-based localization. In Proceedings of the 2017 IEEE international conference on computer vision, Venice, Italy, 22–29 October 2017; pp. 618–626.